Stratified primary care for adults with musculoskeletal pain: the STarT MSK research programme including RCTs

Nadine E Foster,^{1,2} Kate M Dunn,² Joanne Protheroe,² Jonathan C Hill,^{2*} Martyn Lewis,^{2,3} Benjamin Saunders,² Sue Jowett,⁴ Susie Hennings,³ Paul Campbell,^{2,5} Kieran Bromley,^{2,3} Bernadette Bartlam,^{2,6,7} Opeyemi Babatunde,² Simon Wathall,^{2,3} Raymond Oppong,⁴ Jesse Kigozi⁴ and Adrian Chudyk²

¹STARS Education and Research Alliance, Surgical Treatment and Rehabilitation Service (STARS), The University of Queensland and Metro North Health, Brisbane, QLD, Australia

²Primary Care Centre Versus Arthritis, School of Medicine, Keele University, Keele, UK ³Keele Clinical Trials Unit, School of Medicine, Keele University, Keele, UK

- ⁴Health Economics Unit, Institute of Applied Health, University of Birmingham, Birmingham, UK
- ⁵Department of Research and Innovation, Midlands Partnership NHS Foundation Trust, St George's Hospital, Stafford, UK
- ⁶Family Medicine and Primary Care, Lee Kong Chian School of Medicine, Nanyang Technological University, Singapore
- ⁷Community and Primary Care Research Group, Faculty of Health: Medicine, Dentistry and Human Sciences, University of Plymouth, Plymouth, UK

*Corresponding author j.hill@keele.ac.uk

Declared competing interests of authors: All authors report grant funding from the Versus Arthritis Centre of Excellence. Nadine E Foster reports Clinical Trials Unit funding from the National Institute for Health and Care Research (NIHR) (September 2018 to August 2021), membership of the Primary Care, Community and Preventive Interventions Panel (PCCPI) Topic Identification, Development and Evaluation (TIDE) Panel, NIHR Health Technology Assessment (HTA) Programme (2013–15) and funding support from an NIHR Research Professorship. Sue Jowett reports personal fees from Pfizer (Pfizer Inc., New York, NY, USA) and membership of the HTA Funding Committee Policy Group (February 2019–November 20) and HTA Clinical Evaluation and Trials Committee (November 2016–20).

Published June 2023 DOI: 10.3310/FBVX4177

Plain language summary

Stratified primary care for adults with musculoskeletal pain: the STarT MSK research programme including RCTs

Programme Grants for Applied Research 2023; Vol. 11: No. 4 DOI: 10.3310/FBVX4177

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain language summary

A round 20% of adults consult their general practitioner each year with a musculoskeletal problem. Musculoskeletal problems represent the largest group of long-term conditions for which patients consult their general practitioner. Although many patients have mild symptoms that can be self-managed with support, some have more serious or persistent problems that need treatments such as physiotherapy. Care varies widely because it is not easy to identify who should be supported to self-manage and who needs more treatment.

In this programme we adapted and finalised a short questionnaire [the Keele STarT MSK (Subgrouping for Targeted Treatment for Musculoskeletal pain) Tool] to sort patients consulting their general practitioner with back, neck, shoulder, knee or multisite pain into one of three groups: those at low, medium and high risk of a poor outcome (for example, patients still experiencing high levels of pain or disability at 6-month follow-up). We recommended treatment options that general practitioners could provide, or offer referral to, for each group of patients and gave general practitioners training to use the new approach. We consulted patients, general practitioners and physiotherapists about this new approach and tested whether or not it led to better outcomes for patients and whether or not it was good value for money for the NHS.

Results showed that the Keele STarT MSK Tool works well to identify patients at low, medium and high risk of poor outcome. The approach had several positive impacts on general practitioner clinical decision-making and received positive feedback from patients and clinicians. However, matching groups of patients to the available treatments recommended in this programme did not lead to consistently better patient outcomes than those of patients receiving usual care. We found stratified care had greater potential for the group of patients at high risk.

The challenge remains to improve primary care treatments in ways that lead to better outcomes for patients with musculoskeletal pain.

Programme Grants for Applied Research

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

Programme Grants for Applied Research (PGfAR) was launched in 2013 and is indexed by Europe PMC, NCBI Bookshelf, DOAJ, Ulrichsweb[™] (ProQuest LLC, Ann Arbor, MI, USA) and Scopus® (Elsevier, Amsterdam, Netherlands).

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar.

Criteria for inclusion in the Programme Grants for Applied Research journal Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Programme Grants for Applied Research programme

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health and Care Research (NIHR), was established in 2006 to fund collaborative, multidisciplinary programmes of applied research to solve health and social care challenges. Findings are expected to provide evidence that lead to clear and identifiable patient benefits, in the relatively near future.

PGfAR is researcher led and does not specify topics for research; however, the research must be in an area of priority or need for the NHS and the social care sector of the Department of Health and Social Care, with particular emphasis on health and social care areas that cause significant burden, where other research funders may not be focused, or where insufficient funding is available.

The programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director. For more information about the PGfAR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/programme-grants-for-applied-research.htm

This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-1211-20010. The contractual start date was in June 2014. The final report began editorial review in February 2021 and was accepted for publication in June 2022. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, PGfAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the PGfAR programme or the Department of Health and Social Care.

Copyright © 2023 Foster *et al.* This work was produced by Foster *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/.For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress, final files produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).

NIHR Journals Library Editor-in-Chief

Dr Cat Chatfield Director of Health Services Research UK

NIHR Journals Library Editors

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editorin-Chief of HSDR, PGfAR, PHR journals

Dr Peter Davidson Interim Chair of HTA and EME Editorial Board. Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Rob Riemsma Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk